Author's response to reviews

Title: Prevalence and distribution of high-risk Human papilloma virus (HPV) types in invasive squamous cell carcinoma of the cervix and in normal women in Andhra Pradesh, India

Authors:

A Pavani Sowjanay (pavani@cdfd.org.in)
Meenakshi Jain (docmeena@rediffmail.com)
Usha RANI Poli (usharani@eth.net)
Samavedam Padma (padma@cdfd.org.in)
Manik Das (das@yahoo.com)
Keerti V Shah (kvshah@ihsph.edu)
B N Rao (bachin_rao@yahoo.com)
Radha RAMA Devi (radha@cdfd.org.in)
Patti E Gravitt (pgravitt@jhspsh.edu)
Gayatri Ramakrishna (gayatri@cdfd.org.in)

Version: 4 Date: 24 September 2005

Author's response to reviews:

To, 24.9.05

The Editor,
BMC Infectious Disease,

Subject: Revision of manuscript # Manuscript ID 166681037723737

Dear Editor,

As per the reviewers comment, we are submitting the revised manuscript. The details of the changes made is provided in the accompanying document "answers". The queries raised by the reviewers have been answered point wise and the corresponding changes made in the manuscript have been given in parenthesis, with page and line number, in the accompanying answers. We have now removed Fig.1 as suggested by the reviewer.

Thanking you and look forward to hearing from you.

Gayatri Ramakrishna